Author's response to reviews

Title: Are the elderly more vulnerable to psychological impact of natural disaster? A population-based survey of adult survivors of the 2008 Sichuan earthquake

Authors:

Zhaobao Jia (zhaobao.jia@smmu.edu.cn)
Wenhua Tian (wh_tian@smmu.edu.cn)
Weizhi Liu (liuweizhi@21cn.com)
Yang Cao (cao@smmu.edu.cn)
Jin Yan (jinyan@smmu.edu.cn)
Zhisheng Shun (mxcdcbgse@163.com)

Version: 6 Date: 24 March 2010

Author's response to reviews:

Dear Editor,

We have made 3 minor revisions:
1. we updated the order of the authorship according to the contribution to the article.
2. The word "UD$" in the first paragraph(Page 5) of the "Methods" section was replaced by "USD".
3. The words "except for" in the "Data Collection" section (Page 7) were replaced by "except".

with best regards,
Wenhua Tian